Author's response to reviews

Title: Attitudes and perceptions of nephrology nurses towards dialysis modality selection: a survey study

Authors:

Karthik K Tennankore (ktennankore@gmail.com)
Jay Hingwala (jayhingwala@gmail.com)
Diane Watson (Diane.Watson@uhn.ca)
Joanne M Bargman (Joanne.Bargman@uhn.ca)
Christopher T Chan (Christopher.Chan@uhn.ca)

Version: 2 Date: 10 June 2013

Author's response to reviews:

Dear BMC,

Thank you for the pre-review of our manuscript; Attitudes and perceptions of nephrology nurses towards dialysis modality selection: a survey study (MS 8795801491011500. We have uploaded a modified manuscript with the following change:

1. We provided a weblink to the survey in the initial sentence of the methods section:

The survey was created within surveymonkey.net online software using a modified Delphi process (web link to survey: https://www.surveymonkey.com/s/QBVY7TD ).

Thanks you,

Karthik Tennankore